Author’s response to reviews

Title: LEARN 2 MOVE 16-24: Effectiveness of an intervention to stimulate physical activity and improve physical fitness of adolescents and young adults with spastic cerebral palsy; a randomized controlled trial

Authors:

Jorrit Slaman (j.slaman@erasusmc.nl)
Marij Roebroeck (m.roebroeck@erasusmc.nl)
Jetty van Meeteren (j.vanmeeteren@erasusmc.nl)
Wilma van der Slot (w.vanderslot@erasusmc.nl)
Heleen Reinders-Messink (H.A.Reinders@revalidatie-friesland.nl)
Eline Lindeman (e.lindeman@umcutrecht.nl)
Henk Stam (h.j.stam@erasusmc.nl)
Rita van den Berg-Emons (h.j.g.vandenberg@erasusmc.nl)

Version: 5 Date: 4 November 2010

Author’s response to reviews: see over
Dear dr. Aldcroft,

I have made some changes to the manuscript following your request:

**Major**
- Reference 23 has been accepted in the meanwhile. I have changed this in the reference list.
- Reference 24 has been accepted in the meanwhile. I have changed this in the reference list.
- Reference 25 has been accepted in the meanwhile. I have changed this in the reference list.
- I have deleted the citations from figure one and added a paragraph on page 11 to describe the secondary outcome measures.

**Minor**
- I have removed the line numbers.
- I have removed the vertical lines from the tables.
- I have restyled the tables without merged elements

Best regards,

Jorrit Slaman